# Platelet-rich plasma injection for adults with acute Achilles tendon rupture: the PATH-2 RCT

Joseph Alsousou,<sup>1\*†</sup> David J Keene,<sup>2†</sup> Paul Harrison,<sup>3</sup> Philippa Hulley,<sup>4</sup> Susan Wagland,<sup>2</sup> Jacqueline Y Thompson,<sup>2</sup> Scott R Parsons,<sup>2</sup> Christopher Byrne,<sup>5</sup> Michael M Schlüssel,<sup>6</sup> Heather M O'Connor,<sup>6</sup> Susan J Dutton,<sup>6</sup> Sarah E Lamb<sup>2,6</sup> and Keith Willett<sup>2</sup> on behalf of the PATH-2 study group

<sup>1</sup>Institute of Translational Medicine, University of Liverpool, Liverpool, UK <sup>2</sup>Kadoorie Centre for Critical Care Research, John Radcliffe Hospital, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK

- <sup>3</sup>Institute of Inflammation and Ageing, College of Medical and Dental Sciences, University of Birmingham Medical School, University of Birmingham, Birmingham, UK
- <sup>4</sup>Botnar Research Centre, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK
- <sup>5</sup>School of Health Professions, Faculty of Health and Human Sciences, University of Plymouth, Plymouth, UK
- <sup>6</sup>Oxford Clinical Trials Research Unit, Centre for Statistics in Medicine, Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK

\*Corresponding author JosephAlsousou@doctors.org.uk †Joint first authors

Declared competing interests of authors: Joseph Alsousou reports a grant from the National Institute for Health Research (NIHR) Academic Clinical Fellowship programme during the conduct of the study. David J Keene reports grants from a NIHR postdoctoral fellowship during the conduct of the study. Jacqueline Y Thompson reports that she was in receipt of a grant from Thames Valley Clinical Research Network Contingency Fund for physiotherapist training. Sarah E Lamb reports grants from the NIHR Health Technology Assessment (HTA) programme for the AIM study during the conduct of this study. Sarah E Lamb was also a member of the HTA Additional Capacity Funding Board (ongoing), HTA Clinical Trials Board (2010–15), HTA End of Life Care and Add-on Studies (2015), HTA Funding Boards Policy Group (formerly Clinical Study Group) (2010–15), HTA Maternal, Neonatal and Child Health Methods Group (2013–15), HTA post-board funding teleconference (2010–15), HTA Primary Care Themed Call Board (2013–14), HTA Prioritisation Group (ongoing) and NIHR Clinical Trials Unit Standing Advisory Committee (2012–16) during this study. Sarah E Lamb is Co-director of the Oxford Clinical Trials Research Unit and Professor of Rehabilitation at Warwick Clinical Trials Unit, receiving funding from NIHR for both. Sarah E Lamb also receives funding from the NIHR Collaboration for Leadership in Applied Health Research and Care Oxford at Oxford Health NHS Foundation Trust. Keith Willett declares design royalties from Zimmer Biomet (Warsaw, IN, USA), outside the submitted work, for intramedullary bone fixation implants, and held the role of Medical Director for Acute Care, NHS England, during the conduct of this study.

Published November 2019 DOI: 10.3310/eme06120

# **Scientific summary**

The PATH-2 RCT Efficacy and Mechanism Evaluation 2019; Vol. 6: No. 12 DOI: 10.3310/eme06120

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Scientific summary**

# Background

Achilles tendon rupture (ATR) accounts for 20% of all tendon ruptures, and leads to significant health-care and societal costs. The current treatment strategies are (1) surgical repair or (2) immobilisation in a cast or boot. The mechanical and biological properties of healed tendons appear to never match those of the original intact tendons, leading to a high risk of re-rupture (3–5%) or reduced function and a loss of, on average, 63–108 days of work.

Platelet-rich plasma (PRP) is an autologous, supraphysiological concentration of platelets that also contains other blood cells. Platelets play an important role at various stages of the repair process of tendon injury. On activation, platelets release an ordered sequence of growth factors, cytokines and an array of bioactive proteins over the lifespan of the platelets. Subsequently, this leads to recruitment of leucocytes, local stem cells and tenocytes to initiate the healing process. Different methods of PRP preparation result in biological component variability, which may influence its efficacy.

In published studies, there is substantial variation in the validity and type of outcomes measured, as well as inconsistency in the observed effect size of PRP. The underpowered and inadequately designed studies suggest that no definite conclusions can be made on PRP application as an adjunct to standard care in the management of ATR. Prior to the PRP in Achilles Tendon Healing (PATH-2) trial, the authors of a meta-analysis of PRP for orthopaedic conditions concluded that there was a need for adequately powered studies using disease-specific and patient-important outcomes to investigate the effect of PRP (Sadoghi P, Rosso C, Valderrabano V, Leithner A, Vavken P. The role of platelets in the treatment of Achilles tendon injuries. *J Orthop Res* 2013;**31**:111–18).

# **Objectives**

- To evaluate the clinical efficacy of PRP among patients with acute ATR using an objective measure of mechanical muscle-tendon function.
- To evaluate the secondary outcome measures of patient-reported function, pain, participant goal attainment and quality of life.
- To determine the key components of PRP that may contribute to its mechanism of action.
- To identify the tissue-level parameters that PRP may alter to exert its effects in an exploratory biopsy substudy.

# **Methods**

#### Design

A multicentre, parallel-group, participant- and outcome assessor-blinded randomised controlled trial comparing PRP with a placebo (imitation) injection in adults with acute ATR. Two substudies were embedded in the main study to contribute to the understanding of the PRP mechanism in tendon healing:

- substudy 1 PRP and whole-blood analysis
- substudy 2 immunohistochemistry analysis of ultrasound-guided needle biopsies from 16 participants at one centre (Oxford).

#### Setting

The trial was conducted in the trauma and orthopaedic surgery departments of 19 NHS hospitals in England and Wales.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Alsousou et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: INIRH Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

#### **Participants**

Patients aged  $\geq$  18 years with an acute ATR attending an outpatient trauma or orthopaedic clinic within 12 days of sustaining the injury and suitable for non-surgical management were eligible for the trial.

The following patients were excluded:

- those with insertional or musculotendinous junction rupture
- those with previous tendon or ankle injury
- those with deformity to either lower leg
- those with a history of diabetes mellitus
- those with known platelet or haematological disorder
- those using systemic cortisone or anticoagulant treatment
- those with lower-limb gangrene/ulcers or peripheral vascular disease or hepatic or renal impairment
- pregnant or breastfeeding females
- those receiving radiotherapy or chemotherapy
- those with inadequate venous access
- those unable to participate in the trial or attend follow-up.

#### Interventions

Participants were individually randomised to receive either PRP injection or placebo (dry-needle insertion to the tendon rupture gap), preceded by local anaesthetic, in a 1 : 1 allocation ratio. A central computer-based randomisation system utilising minimisation, stratified by centre and age group (< 55 years or  $\geq$  55 years), with a probabilistic element of 0.8 to reduce predictability, was provided by the Oxford Clinical Trials Research Unit. Immediately after randomisation, up to 55 ml of venous blood was taken from participants in the PRP group and up to 5 ml was taken in the placebo group. Both interventions were delivered using the same technique by a surgeon or extended-scope physiotherapist while maintaining a participant's blinding. Post injection, the remaining blood and PRP samples were sent to a central laboratory for substudy 1 analysis. Sixteen participants (nine in the PRP group and seven in the placebo group) in one centre (Oxford) received an ultrasound-guided biopsy for substudy 2 assays. All participants received standardised rehabilitation in terms of the duration of ankle immobilisation and non-weight-bearing, and all were referred for physiotherapy.

#### Follow-up

Blinded outcome assessments were carried out at 4, 7, 13 and 24 weeks post randomisation. Following signed consent being obtained, baseline data were collected and the participant was randomised; in most cases, the injection treatment took place on the same visit. Primary outcome data were collected at a 24-week face-to-face appointment. At every time point, trial follow-up was carried out wherever possible by blinded assessors unaware of treatment allocation.

#### **Outcome measures**

Muscle–tendon function assessed by the Limb Symmetry Index (LSI) of work (joules) during the heel-rise endurance test (HRET) at 24 weeks was the primary outcome. Movement of the heel during the HRET in each leg was captured using a computer-controlled linear encoder. The work LSI was calculated as follows: (injured limb measurement/uninjured limb measurement) × 100.

Secondary outcomes were the maximum heel-rise height and number of repetitions during the HRET and the patient-reported outcomes of function and symptoms [measured using the Achilles tendon Total Rupture Score (ATRS)], quality of life [measured using the Short Form questionnaire-12 items (SF-12) version 2 acute], pain (measured using the visual analogue scale and subscale from ATRS) and participant goal attainment [measured using the Patient-Specific Functional Scale (PSFS)].

In substudy 1, whole-blood and PRP samples were analysed for cell count, platelet activation and growth factor concentrations [i.e. platelet-derived growth factor (PDGF), insulin-like growth factor 1 (IGF-1), vascular endothelial growth factor (VEGF), fibroblast growth factor (FGF) and transforming growth factor beta (TGF- $\beta$ )]. In substudy 2, 16 participants in one centre had needle biopsy under ultrasound guidance

at 6 weeks. Analysis included tissue morphology, proliferation, apoptosis, vascularity, metabolic indicators and collagen ratio.

#### Analysis

The target sample size was 230 participants to provide 80% power. For the primary outcome, analysis included a modified intention-to-treat (mITT) population, defined as all randomised intention-to-treat participants with available work LSI data. Multivariate linear regression was used to investigate the effect of PRP on ATR recovery. Sensitivity analyses were carried out using imputation of values for missing HRET data to examine the robustness of the conclusions made from the analyses to address the primary aims of the trial.

A mITT population was also used for secondary outcome analyses. Linear mixed-effects regression models were used to allow the data collected at all follow-up time points to be taken into account, adjusting for pre-injury baseline scores when applicable. Data quality and effect of treatment received were assessed using complier-average causal effect (CACE) analysis in place of the originally planned per-protocol analysis. Complication events reported by participants were explored at two levels: serious adverse events and adverse events (AEs).

For the two substudies, analyses were primarily descriptive, and the relationship between various biomarkers and clinical outcomes was explored.

#### Results

A total of 230 participants were recruited between July 2015 and September 2017. Of these, 114 were randomised to receive the PRP injection and 103 (90%) of these received the allocated treatment; 116 were allocated to, and received, placebo. At 24 weeks, 201 out of 230 participants (87.4%) completed the HRET to provide the work LSI primary outcome, and 216 out of 230 (93.9%) completed the patient-reported outcomes. One participant withdrew from the trial. The average age of participants was 45 years; 75% were male, with 69% of injuries occurring during sporting activity. The baseline characteristics of the participants in the intervention groups were well matched.

#### Clinical trial results

There was no difference between the PRP and placebo groups at 24 weeks in the work LSI. In the PRP group (n = 101), the work LSI was 34.9%, compared with 38.3% in the placebo group (n = 100) [adjusted mean difference -3.872, 95% confidence interval (CI) -10.454 to 2.710; p = 0.231]. Statistical model adjustment by stratification factors and the predefined prognostic variables had no impact on the results attained. Sensitivity analyses accounting for participants with zero measurements for the uninjured limb (unable to lift the heel at all) in the HRET, individuals missing heel-rise repetitions, individuals missing the entire HRET data sets and compliance (i.e. CACE) showed that the results were robust.

There was no difference in secondary outcome results at 24 weeks: ATRS [PRP (n = 107), mean 64.9; placebo (n = 109), mean 65.6; adjusted mean difference -0.543; 95% CI -4.899 to 3.813; p = 0.807] and PSFS (PRP, n = 109, mean 7.198; placebo, n = 107, mean 7.495; adjusted mean difference -0.297; 95% CI -0.868 to 0.274; p = 0.291). ATRS-related pain scores were not different between the two groups in the follow-up period (PRP, n = 109, mean 7.661; placebo, n = 107, mean 7.449; adjusted mean difference 0.212; 95% CI -0.563 to 0.987; p = 0.592). Although no differences in the SF-12 physical component score were identified between the treatment groups (adjusted mean difference 0.805, 95% CI -1.269 to 2.879; p = 0.447), mean SF-12 mental component scores were lower in the PRP group than in the placebo group at 24 weeks (adjusted mean difference -2.714, 95% CI -5.242 to -0.187; p = 0.035). There was no difference between the PRP group and the placebo group in any of the patient-reported secondary outcomes at 4, 7 and 13 weeks. Daily pain over the 2 weeks after injection was not difference -4.019; 95% CI -10.302 to 2.265; p = 0.210). The two groups had similar AE rates related to their Achilles rupture or injection. The number of participants reporting at least one complication of any type related to their Achilles rupture or injection was 84 out of

<sup>©</sup> Queen's Printer and Controller of HMSO 2019. This work was produced by Alsousou *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIRR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

113 (74%) for the PRP group and 90 out of 116 (78%) for the placebo group. The numbers of participants experiencing a re-rupture [PRP, 6/113 (5.3%); placebo, 4/116 (3.5%)] and deep-vein thrombosis [PRP, 6/113 (5.3%); placebo, 5/116 (4.3%)] were also similar.

#### Substudy 1 results

Whole-blood cell counts (red blood cells, white blood cells and platelets) showed that the two groups were relatively well matched at baseline. Cell count analysis of PRP samples showed wide variation in cell counts. The mean platelet count was  $852.6 \times 10^{9}/I$  [standard deviation (SD)  $439.0 \times 10^{9}/I$ ], with a wide range from 6.0 to  $2903.0 \times 10^{9}/I$ . The mean white blood cell count was  $15.1 \times 10^{9}/I$  (SD  $10.3 \times 10^{9}/I$ ), with a range of 1.7 to  $65.3 \times 10^{9}/I$ . Red blood cells were reduced remarkably ( $0.9 \times 10^{12}/I$ , SD  $1.5 \times 10^{12}/I$ , range 0.1 to  $9.0 \times 10^{12}/I$ ). The quality of the PRP samples in the majority of preparations was high, with low levels of basal activation, and they were capable of activation and degranulation. TGF- $\beta$ , VEGF, PDGF, IGF-1 and FGF mean concentrations (133.4 ng/mI, 0.984 ng/mI, 55.49 ng/mI, 78.2 ng/mI and 112.5 pg/mI, respectively) were high, as expected. Overall, PRP samples were therefore shown to be functional, with the majority of platelets in the PRP preparations shown to be capable of activation and degranulation. Parameters of baseline whole blood taken before intervention in both groups did not correlate with the primary outcome measure at 24 weeks. PRP cell counts did not correlate with the primary outcome measure. None of the growth factor concentrations showed any correlation with the work LSI.

#### Substudy 2 results

All biopsy results except one showed evidence of healing at 6 weeks; collagen fibre density was lower in the PRP group. This did not correlate with differences in cellularity or vascularity as these parameters were similar in both groups, suggesting equivalent healing processes.

## Conclusions

#### Implications for health care

The main finding of the PATH-2 trial is that there was no evidence of benefit for PRP application in acute ATR in terms of objective and subjective efficacy outcomes. The effect size estimates of the primary outcome and end point and the consistency with patient-reported secondary outcomes during the follow-up strongly support the validity of the conclusion that PRP does not improve the outcome of ATR management. Although a health economic analysis was not carried out, applying PRP in ATR management would add to the cost of standard care for no clinically measured improvement in the outcome. It is a no-value intervention in ATR management.

#### **Recommendations for research**

The implication of the PATH-2 trial is that the indication for PRP application in other soft-tissue injuries should be validated by similar robust clinical trials. The extent of functional asymmetry between injured and uninjured legs in this trial was substantial. Optimising recovery of tendon–muscle function during rehabilitation is therefore a recommended area of future investigation. An extended follow-up of PATH-2 participants at 2 years has started to evaluate longer-term patient-reported outcomes.

## **Trial registration**

This trial is registered as ISRCTN54992179.

## Funding

This project was funded by the Efficacy and Mechanism Evaluation programme, a Medical Research Council and National Institute for Health Research (NIHR) partnership. The trial was supported by the NIHR Biomedical Research Centre, Oxford, and the NIHR Fellowship programme.

# **Efficacy and Mechanism Evaluation**

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme support translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

#### This report

The research reported in this issue of the journal was funded by the EME programme as project number 12/206/30. The contractual start date was in October 2014. The final report began editorial review in September 2018 and was accepted for publication in March 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Alsousou *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

# **NIHR Journals Library Editor-in-Chief**

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk